Author's response to reviews

Title: A qualitative study of factors affecting mental health amongst low-income working mothers in Bangalore, India.

Authors:

Sandra M Travasso (sandramarytravasso@gmail.com)
Divya Rajaraman (Divya.rajaraman@sjri.res.in)
S Jody Heymann (jody.heymann@ph.ucla.edu)

Version: 2 Date: 26 June 2013

Author's response to reviews:

The revised manuscript contains formatting changes which involved including author details on the tile page and including abstract, author contributions and acknowledgements within the main manuscript.